\begin{table}%t7
 %\centering
\par
\caption{\label{tab_magnitude}Photometry of GRB host galaxies.}
\par
\begin{tabular}{lccccccccccc}
\hline\hline \noalign{\smallskip}
 {GRB} & {$U$} & {$B$} & {$V$} & {$R$} & {$I$} & {$J$} &
 {$H$} & {$K$} & {Refs.} & \\
\hline
980703  &  -- & $23.40 \pm 0.12$  &  $23.04 \pm 0.08$  &  $22.58 \pm 0.06$  &  $21.95 \pm 0.25$ & $20.87 \pm 0.11$ & $20.27 \pm 0.19$ & $19.62 \pm 0.12$ & 1, 2, 3 \\
990712  &  $23.12 \pm 0.05$ & $23.36 \pm 0.09$ & $22.39 \pm 0.03$ &  $21.84 \pm 0.02$ &  $21.41 \pm 0.03$  &  $20.81 \pm 0.17$ &  $20.25 \pm 0.19$  &  $20.05 \pm 0.10$  &  3   \\
020903  &  --  &  $21.70 \pm 0.10$  &  $20.80 \pm 0.10$ & $20.80 \pm 0.10$ &  $20.50 \pm 0.10$  &  --  &  --  & --  &    4, 5    \\
030329  &  $22.68 \pm 0.10$ & $23.42 \pm 0.07$ & $22.88 \pm 0.05$  &  $22.80 \pm 0.04$ & --  & $21.52 \pm 0.04$  &  $21.17 \pm 0.24$  &  --   &  6  \\
031203  &  $22.41 \pm 0.18$  &  $22.32 \pm 0.05$  & $20.53 \pm 0.05$ &  $20.44 \pm 0.02$ & $19.40 \pm 0.04$  &  $18.28 \pm 0.02$ &  $17.78\pm0.02$  &  $16.54 \pm 0.02$  &   7  \\
060218  &  $20.45 \pm 0.15$  & $20.46 \pm 0.07$  & $20.19 \pm 0.04$    &  19.86~$\pm$~0.03     &  19.47~$\pm$~0.06     &  18.99~$\pm$~0.16    &  18.52~$\pm$~0.22  &  18.69~$\pm$~0.34   &  8, 9, 10 \\
060505  &  18.43~$\pm$~0.05  &  18.89~$\pm$~0.02  &  18.27~$\pm$~0.02  &  17.90~$\pm$~0.02  &  17.51~$\pm$~0.02  &  17.29~$\pm$~0.08$^a$  &  --  &  15.85~$\pm$~0.04  &  11  \\
\noalign{\smallskip}\hline
  & -- & -- & F555W$^b$ & F702W$^b$ & F814W$^b$ & -- & -- & -- & \\
020405 & -- & -- & 22.63~$\pm$~0.05 & 21.84~$\pm$~0.05 & 21.29~$\pm$~0.05 & -- & -- & -- & 12 \\
\hline
\end{tabular}
\tablebib{(1)  \citet{Vreeswijk99}; (2) \citet{Sokolov01}; (3) \citet{Christensen04};  (4) \citet{Soderberg04};
(5) \citet{Bersier06};  (6) \citet{Gorosabel05}; (7) \citet{Margutti07}; (8) \citet{Sollerman06};
(9) \citet{Castro08};
(10) \citet{Kocevski07}; (11) \citet{Thone08}; (12) \citet{Wainwright07}.}\vspace{-2mm}
\tablefoot{The magnitudes are not corrected for Galactic extinction. 
\tablefoottext{a}{FORS1 $z$ filter (Central wavelength: 9100~\AA; band width: 1305~\AA).} 
\tablefoottext{b}{F555W (Central wavelength: 5407~\AA; band width: 1236~\AA); F702W (Central wavelength: 6895~\AA; band width: 1389~\AA);
F814W (Central wavelength: 7940~\AA; band width: 1531~\AA)}.}\vspace{2mm}
\end{table}